Author’s response to reviews

Title: Advancing the Application, Quality and Harmonization of Implementation Science Measures

Authors:

Adrienn B Rabin (borsika.a.rabin@gmail.com)
Peyton Purcell (purcellp@mail.nih.gov)
Sana Naveed (sana.naveed@nih.gov)
Richard P Moser (moserr@mail.nih.gov)
Michelle D Henton (michelle.d.henton@kp.org)
Enola K Proctor (ekp@wustl.edu)
Ross C Brownson (rbrownson@wustl.edu)
Russell E Glasgow (glasgowre@mail.nih.gov)

Version: 4 Date: 17 October 2012

Author’s response to reviews: see over
October 17, 2012

Dr. Gregory Aarons
Editor
Implementation Science

RE: Revision of manuscript 1359409465748507 “Advancing the Application, Quality and Harmonization of Implementation Science Measures”

Thank you for the review and acceptance of our manuscript in Implementation Science. In the attachment, we have provided response to the last set of review comments. We have also made revisions in the manuscript based on the review comments and indicated changes using track changes.

All authors of this research paper have directly participated in the planning, execution, or analysis of the study. All authors of this paper have read and approved the final version submitted. The contents of this manuscript have not been copyrighted or published previously. The contents of this manuscript are not now under consideration for publication elsewhere. The contents of this manuscript will not be copyrighted, submitted, or published elsewhere while acceptance by the Journal is under consideration. There are no directly related manuscripts or abstracts, published or unpublished, by any author(s) of this paper.

We greatly appreciate your help with moving the manuscript through the review process in a timely manner.

Sincerely,

Borsika A. Rabin, M.P.H., Ph.D.
Staff Researcher
CRN Cancer Communication Research Center
Institute for Health Research
Kaiser Permanente Colorado
Phone: (303) 614-1295
Regular mail: P.O. Box 378066, Denver CO 80237 USA
E-mail: borsika.a.rabin@gmail.com

Russ Glasgow, PhD
Deputy Director, Implementation Science
Division of Cancer Control and Population Sciences, National Cancer Institute
Responses to Reviewers’ comments:

Reviewer: Cara Lewis

Major Compulsory Revisions
None

Minor Essential Revisions
The revised manuscript would benefit from a careful read to address grammatical errors. The following are some of the identified errors: an extra space in the 3rd sentence on page 8; an extra comma in the 3rd to last sentence on page 12; an extra period in the last sentence of the 2nd paragraph as well as the 2nd sentence of the 3rd paragraph on page 18; an extra word (“to”) in the last sentence of the 3rd paragraph on page 18; incorrect verb ending (adding versus add) in 1st sentence of 2nd paragraph on page 20.

Thank you for the thorough review and for pointing out these errors. We have reviewed the entire manuscript and fixed the above-mentioned and additional errors we found.

Discretionary Revisions
The authors might consider breaking up their description of the 4-step process in the introduction on page 7 (e.g., First, … Second, …) as opposed to including all details in a single sentence using a numbered list as there is simply a lot of information to digest.

We have modified this sentence as suggested above.